Author’s response to reviews

Title: Prevalence and clinical associations of wheezes and crackles in the general population. The Tromsø Study

Authors:
Juan Carlos Aviles (juan.c.solis@uit.no)
Cristina Jácome (cristinajacome.ft@gmail.com)
Anne Davidsen (annehdav@gmail.com)
Raimonda Einarsen (raimonda.einarsen@gmail.com)
Sophie Vanbelle (sophie.vanbelle@maastrichtuniversity.nl)
Hans Pasterkamp (hpasterkamp@shaw.ca)
Hasse Melbye (hasse.melbye@uit.no)

Version: 3 Date: 06 Aug 2019

Author’s response to reviews:

Dear Editor:

Thanks again for considering our publication for publication.

About your remark on written consent for publication.

As I wrote in the past rebuttal letter I can confirm once again that the individual in the picture of figure 1 is the corresponding author (meaning myself). I have filled out the consent form and included it in the submission as requested.

I have also changed the manuscript declaration in the section consent to publish which now reads:

"Written informed consent was obtained from the patient for publication of Figure 1. A copy of the written consent is available for review by the Editor of this journal."

Thank you in advance

best regards
Dr. Juan Carlos Aviles Solis